Author’s response to reviews

Title: Invasive cutaneous Neoscytalidium infections in renal transplant recipients: a series of five cases

Authors:

Simon Garinet (simon.garinet@orange.fr)
Jerome Tourret (jerome.tourret@aphp.fr)
Stephane Barete (stephane.barete@aphp.fr)
Nadia Arzouk (nadia.arzouk@aphp.fr)
Isabelle Meyer (ismayameyer@gmail.com)
Camille Frances (camille.frances@aphp.Fr)
Annick Datry (annick.datry@aphp.fr)
Dominique Mazier (dominique.mazier@aphp.Fr)
Benoit Barrou (benoit.barrou@aphp.fr)
Arnaud Fekkar (arnaud.fekkar@psl.aphp.fr)

Version: 3 Date: 24 Sep 2015

Author’s response to reviews:

Editorial Requests:

1. Keywords:

Please include three to ten keywords representing the main content of the article.

>>The following keywords were present just before the abstract (page 2): Fungal infection; Renal transplant; Solid organ transplant; Immunodepression; Moulds. We have moved them after the abstract.

2. Consent:

Please clarify in the consent statement on p17 that all patients gave consent for publication. Currently the consent statement only refers to one patient.

>>Page 12. We have changed the sentence.
3. Re-name 'CONTRIBUTIONS' to 'Authors Contributions'.

>>Done.

4. Abbreviations:

Please include a list of abbreviations used in the manuscript and their meanings after the Conclusions section.

>>Done.

5. Figure Legends:

Please remove the Figure legends in the figure files. They should be included within the main manuscript.

>>We have removed the figure legends in the figure files. They were already included after the references.